RESEARCH Open Access

Contextual-relationship and stress-related factors of postpartum depression symptoms in nulliparas: a prospective study from Ljubljana, Slovenia



Polona Rus Prelog^{1*}, Marijana Vidmar Šimic², Tanja Premru Sršen^{2,3} and Maja Rus Makovec^{1,3}

Abstract

Background: For a significant proportion of women, postpartum depression (PPD) is the first mood episode in their lives, yet its aetiology still remains unclear. Insecure attachment in close adult relationships is considered to be a risk factor for depressive symptoms. This study aimed to gain further insight into the risk factors for postpartum depression symptoms (PPDS) of nulliparas in Slovenia and to examine vulnerability to developing depressive symptoms, with an emphasis on contextual and stress-related characteristics.

Methods: The sample consisted of 156 nulliparas in the third trimester of pregnancy enrolled in a childbirth preparation program. The following instruments were applied: Experiences in Close Relationships-Revised, the Edinburgh Postpartum Depression Scale (EPDS), the Zung Anxiety Scale and a question battery designed by the research team including questions about emotional support and work-related stress. Logistic regression was used to test the association between demographic, social, environmental, personality and attachment variables and PPD of nulliparas (EPDS ≥10), controlling for baseline (prepartum) depression score. A multivariable linear regression model was built with the postpartum EPDS continuous score as a dependent variable.

Results: 28/156 (17,9%) were evaluated as being at risk for depression (EPDS≥10) in the last trimester and 25/156 (16%) at six weeks postpartum. The results of the logistic regression model controlled for prepartum depression score showed that increased risk for developing PPDS was associated with anxiety level postpartum, intimate-partner-attachment anxiety postpartum, and elevated stress due to loss of employment or an unsuccessful search for employment in the previous year. The results of the multivariable regression model, however, showed the association with education and postpartum anxiety with PPDS continuous score; EPDS after giving birth was higher for more educated and more anxious primiparas.

Conclusions: Our findings demonstrate the importance of anxiety symptoms and higher education level in assessments of nulliparas' mental health. The results of our study show and confirm the results of previous research that anxiety symptoms in the immediate postpartum period are likely to be associated with depressive symptoms in nulliparas. The results also suggest that higher level of education of first-time mothers might not be a protective factor, especially for nulliparas with the university level of education. Further studies on larger samples should be considered.

Keywords: Nulliparas, Postpartum depression, Anxiety, Education, Partner attachment

^{*} Correspondence: ruspolona@gmail.com; polona.rus@psih-klinika.si

1 University Psychiatric Clinic Ljubljana, Studenec 48, 1000 Ljubljana, Slovenia
Full list of author information is available at the end of the article



Plain English summary

Depression in women after childbirth (postpartum depression - PPD) is a well-known health-care problem and a serious economic burden. However, it is still often diagnosed late although half of mood disorders actually begin before delivery, in the last trimester of pregnancy.

In recent years, several risk factors have been identified. Those with the strongest impact are previous depressive episodes, previous PPD, anxiety and depressive disorders during pregnancy, which can be predisposed by life stressors and a lack of social support. An intimate relationship and partner attachment style are also considered important. Pregnancy and transition to motherhood can generate stress which can cause insecurely attached women to be more vulnerable to the development of depression.

In this study we further explored the risk factors for depression symptoms of first-time pregnant women. The sample consisted of 156 women in the third trimester in a childbirth preparation program. They answered a questionnaire when 30 weeks pregnant and six weeks after delivery. We found that 17,9% were at risk for depression before and 16% at six weeks after delivery. Increased risk was associated with postpartum anxiety symptoms and higher education of nulliparas. Intimate-partner-attachment anxiety postpartum, and elevated stress due to loss of employment or an unsuccessful search for employment in the previous year were also found important but were not confirmed in the final statistical model. Therefore, we suggest paying attention to anxiety symptoms of first-time mothers. Our results also suggest that higher education of first-time mothers might not be a protective factor for postpartum depression.

Background

Depression in the postpartum period is a well-known health-care problem and a serious economic burden. For a significant proportion of women, postpartum depression (PPD) is the first mood episode in their lives [1], yet it still often goes unrecognized or is diagnosed late [2, 3]. It has been established that 50% of postpartum mood disorders actually begin prior to delivery and some studies have even shown a higher prevalence of antepartum depression than PPD [4, 5].

Among the factors found to have the strongest impact on the development of PPD are previous depressive episodes, previous PPD, anxiety and depressive disorders during pregnancy [6, 7]. Other predictive factors include a negative birth experience, the mode of delivery (especially Caesarian section), unintended/unplanned pregnancy, lower age and less social support during pregnancy [8, 9].

In recent years, several social, environmental, and stress-related factors have been shown to be associated with PPD. Life stressors and a lack of social support can predispose nulliparas' anxiety and depressive symptomatology. Support from a partner in pregnancy has been shown to have a predictive value for maternal prenatal and postpartum mental health [10] and an intimate relationship and partner attachment style represent an important part of the emotional experience of nulliparas [11].

Insecure attachment in close adult relationships is considered to be a risk factor for depressive symptoms [12]. Studies have found a prospective association between adult attachment and PPD [13] and some have reported attachment anxiety [14] or insecure attachment without specification of subtype or several subtypes [15–17]. An insecure attachment style in adults is thought to be based on less optimal experiences with early caregiving [18, 19], in addition to many other multifactorial influences [20]. Pregnancy and the transition to motherhood can generate stress and specific concerns with a close relationship, which can cause insecurely attached women to be more vulnerable to the development of depression.

Relatively few studies have analysed the working environment and reported work-related distress as risk factors for PPD [21] when considering environmental factors for a predisposition to depression. Gjerdingen [22] found mothers' employment to be consistently associated with lower odds of depressive symptoms at 13 months after delivery. A supportive environment and feeling appreciated at work have been found to be important factors in working throughout pregnancy and the working environment might have a favourable effect on women's health resources [21].

A recent study has found that employment, especially full-time employment and holding a professional or technical job, may reduce the risk of PPD [23]. Workrelated and career questions can represent life stress, especially for women with higher education. In the European Union (EU), the proportion of women aged 30-34 with tertiary education exceeds that of men. Most EU countries have a negative gender gap (defined as the proportion of men aged 30-34 with tertiary education minus that of women), with Slovenia showing among the largest gender gaps in tertiary education attainment (-21.7 p.p.) in Slovenia, compared to -26.0 p.p. in Latvia, which is the largest gender gap in absolute value). In general, women's age at childbirth has been increasing and their rising educational levels feature prominently among the explanations for increasingly later patterns of age at first birth [24].

While it seems to have been established that PPD is most frequent among women with a history of depression, several prospective studies have shown that a majority of women with depression onset in the post-partum period have no prior history of mood disorders [1, 25]. In fact, a recent systematic review has reported a similar prevalence rate of PPD among healthy mothers without a history of depression when compared to mothers with a history of depression.

Parity has also been found important. Studies have shown higher risk of PPD in nulliparas compared to multiparas, especially in the first month postpartum, and higher scores on the EPDS [26]. A recent study also reported higher fear of childbirth in nulliparas, compared to multiparas [27].

To our knowledge, no studies have so far examined factors associated with PPD in nulliparas in Slovenia. The estimated incidence of postpartum depression in Slovenia is 21% [28]. The population structure of nulliparas in Slovenia has been changing in the last years and decades, the age at first birth is rising, as well as the educational level of women at first birth and they also live in multigenerational households longer (SURS). Identifying risk factors for postpartum depressive symptoms in first-time mothers in this population is therefore highly important. The authors of the recent Cochrane review reported that early identification of mothers at high risk for PPD could help to prevent approximately one-third of the cases of PPD [29]. PPD is difficult to predict and, as several authors have noted, evidence on where to focus screening is still lacking [30, 31].

This study aimed to gain further insight into the risk factors for postpartum depression symptoms (PPDS) of first-time pregnant women (nulliparas) and to examine vulnerability to developing depressive symptoms. Beyond the well-researched risk factors (previous psychiatric illness, history of depression, low income, lower education, poor marital relationship, abuse, previous loss of a baby, etc.), we focused our observation on the contextual and stress-related characteristics of nulliparas.

Methods

Study sample

The sample of the present study consisted of 156 pregnant women in the third trimester recruited sequentially from parenting classes held at the University Medical Centre (UMC) Ljubljana's Division of Gynaecology and Obstetrics from March to September 2014. The inclusion criteria were: first pregnancy, third trimester of pregnancy, at least 18 years of age. Excluded from the study were the women who did not fill out the whole EPDS questionnaire at both assessment points.

At baseline, there were 325 women included in the study, of which 181 (55,7%) at least partially completed the questionnaires before and after giving birth. Of these, 156 had complete data on the dependent variable

(EPDS score). These were included in the statistical analysis.

Women that were included in the statistical analysis (n = 156) did not differ from women initially included in the study in any of the variables but in positive experience of birth (p = 0.002) and baseline attachment anxiety (p = 0.012). Women with positive experience of birth and lower baseline attachment anxiety were less likely to drop out of the study.

From 156 women with complete data on EPDS, 90 (57,7%) women were included in the multivariable linear regression model due to missing data and listwise deletion.

A statistically significant difference between missing and non-missing cases was found in the variable measuring stress due to primiparas' physical or mental illness. Women experiencing high stress due to their own physical or mental illness in the last 12 months were less likely to provide complete answers (p = 0.013). There were no statistically significant differences between women included and excluded from the regression model analysis in any other categorical or continuous variable. The demographic characteristics of the sample are listed in Table 1.

Procedure

The study was conducted as a collaboration between obstetricians from UMC Ljubljana's Division of Gynaecology and Obstetrics and psychiatrists from the University Psychiatric Hospital Ljubljana. It was approved by the Republic of Slovenia National Medical Ethics Committee (NMEC) (protocol No. 92/12/13). All of the study participants were given verbal and written explanations of the study and their informed consent was obtained prior to their participation in the study. The study was based on a convenience sample.

The questionnaires were administered during parenting classes from March 2014 to September 2014. The classes are run by midwives and include lectures by a paediatrician, an anaesthesiologist, a dentist, psychologists, and other specialists. The classes are open to pregnant women in their third trimester and are mostly attended by women in their first pregnancy; their partners are welcome to attend. Each class consists of 10 meetings over three weeks. The topics are preparation for labour, birth, and postnatal care of the baby.

The participants completed a structured questionnaire in their third trimester of pregnancy (mean 30 weeks). Each participant was given an anonymous questionnaire with a code, which was saved together with their personal information. During the recruitment period, the midwives and participating doctors invited 696 Slovenian-speaking Caucasian pregnant

Table 1 Characteristics of primiparas (results shown as frequency and percentages if not indicated differently)

	n = 156
Mean age (SD)	30.7 (4.1)
Mean years of education (SD)	16.4 (2.3)
Planned pregnancy	129 (82.7)
Caesarean section ($n = 150$)	32 (21.3)
Positive birth experience ($n = 152$)	126 (82.9)
Multigenerational household ($n = 155$)	33 (21.3)
Constant support from partner ($n = 150$)	131 (87.3)
Constant support from parents ($n = 150$)	94 (62.7)
Constant support from friends ($n = 151$)	66 (43.7)
High support from co-workers b.b. $^{a}(n = 147)$	66 (44.9)
Mean attachment avoidance bb (SD) ($n = 144$)	1.8 (0.7)
Mean attachment anxiety bb (SD) ($n = 141$)	1.7 (0.6)
Mean attachment avoidance ab (SD) ($n = 135$)	1.8 (0.8)
Mean attachment anxiety ab (SD) $(n = 144)$	1.7 (0.7)
Stress - loss of employment ($n = 150$)	15 (10)
Stress – family member or close friend's illness or injury $(n = 151)$	27 (17.9)
Stress - own illness or injury ($n = 149$)	10 (6.7)
Stress - financial difficulties (n = 150)	19 (12.7)
Psychiatric help before pregnancy ($n = 155$)	16 (10.3)
Mean anxiety score bb (SD) (n = 144)	33.2 (5.7)
Mean anxiety score ab (SD) ($n = 150$)	29.6 (5.8)

^abb before giving birth, ab after giving birth

women who attended the classes to take part. Written informed consent to participate was signed by 387 (55.6%) of the women. Participants completed the second questionnaire six weeks after giving birth via the online tool SurveyMonkey. Participants who did not answer the follow up received a maximum of two e-mail reminders. We excluded 38 of the participating women who were in their second or subsequent pregnancies.

A prospective longitudinal design was used. We used SRQR reporting guidelines [32].

Measures/instruments

Partner attachment

Experiences in Close Relationships-Revised (ECR-R) [33]: the ECR-R is a 36-item self-report measure used to assess adult romantic attachment on a 7-point scale ranging from 1 (strongly disagree) to 7 (strongly agree). The scale consists of two 18-item subscales: anxiety (fear of rejection and abandonment) and avoidance (discomfort with closeness and discomfort with depending on others). For our sample, internal consistency was α = .84 both for the Avoidance and the Anxiety scale.

Depressive symptoms

The Edinburgh Postpartum Depression Scale (EPDS) [34]: The EPDS is a self-report questionnaire consisting of 10 items with four ordered response categories scored from 0 to 3. When used as a screening instrument, the cut-off scores of 12/13 usually designate major depression, whereas scores from 9 to 11 indicate mild depression levels in need of further assessment [35]; Cronbach α = .83. We used a cut-off score = 10, considering that this cut-off proved reasonable/appropriate in a previous study using a Slovene sample [36].

Anxiety

The Zung Anxiety Scale [37] consists of 20 items which test the participants' autonomic, motor, cognitive and other anxiety symptoms. For each item the participants choose one of the following answers: a little of the time, some of the time, good part of the time, most of the time; Cronbach $\alpha = .76$.

Sociodemographic and pregnancy information

A question battery designed by the research team included: maternal age; years of education (education level); planned/unplanned pregnancy; living arrangement (shared household with elder generation/own household); mode of delivery (Caesarian section); whether psychiatric help had been received before pregnancy; level of emotional support received from partner, parents, friends and co-workers; stress due to loss of employment/illness/financial problems.

Constant (before and after giving birth) emotional support from partner, parents, friends and emotional support from co-workers during pregnancy was measured on a 5- point self-constructed scale containing 4 questions. The scale that was dichotomized prior to analysis into categories "weak to moderate support" including answers "almost none" to "moderate support" and "strong support" including answers "strong" and "very strong support".

Stress in the last year due to unemployment, family member or close friend's illness, nulliparas' physical or mental illness and financial problems was measured on a 5-point scale that was dichotomised prior to analysis (1 thru 3 = low stress; 4 thru 5 = high stress).

Statistical analysis

Logistic regression was used to test the association between demographic, social, environmental, personality and attachment variables and PPD of nulliparas, controlling for prepartum depression score. PPD was indicated by an EPDS score higher or equal to 10 points. As there was a smaller number of nulliparas with afterbirth depression score above or equal to 10 points (n = 25), no multivariable logistic regression model was built, since

no fewer than 10 cases in the smaller category of the dependent variable per included independent variable is recommended to obtain valid and stable results of the logistic regression model [38, 39]. In the current study up to 2 independent variables could be included in the logistic regression model.

To assess which risk factors play the most important role in postpartum depression, a multivariable linear regression model was built with the continuous after-birth EPDS score as a dependent variable.

The independent variables in each model were: age; years of education; planned pregnancy (yes/no); Caesarean section (yes/no); positive birth experience (yes/no); shared household with elder generation (yes/no); constant (before and after giving birth) emotional support from a partner, parents and friends and emotional support from co-workers during pregnancy (weak to moderate = almost none to moderate support; strong = strong and very strong support); attachment avoidance and attachment anxiety before and after giving birth; stress in the last year due to unemployment or illness of a family member or close friend; nulliparas' physical or mental illness; financial problems (1 through 3 = low stress; 4 through 5 = high stress); psychiatric help before pregnancy (yes/no); Zung's anxiety score before and after giving birth and the EPDS before giving birth. Each of the above-mentioned risk factors was included in the logistic model, controlled for prepartum depression score of nulliparas. Adjusted odds ratios for being at risk for postpartum depression (postpartum EPDS score ≥ 10) were calculated.

The same risk factors were then included in the multivariable linear regression model, where the dependent variable was EPDS postpartum score (continuous variable). There was no threat of multicollinearity as the highest variance inflation factor was 2.9.

P-values < 0.05 (two-tailed) were treated as statistically significant. No adjustment for multiple comparisons was applied. Data were analysed using SPSS version 24 for Windows.

Results

Prevalence of PPD in the study sample

One hundred fifty-six nulliparous women who answered the questionnaire in the last trimester and six weeks postpartum were included in the study. Of those, 28/156 (17,9%) were evaluated as being at risk for depression (EPDS≥10) in the last trimester and 25/156 (16%) at six weeks postpartum.

Socio-demographic characteristics of study subjects

The characteristics of primiparas that completed the questionnaire before and after giving birth are summarized in Table 1.

Results of logistic regression and multivariable linear regression analysis

The results of the logistic regression showed that, when controlled for the prepartum depression score, EPDS after giving birth above or equal to 10 points was associated with higher anxiety score and higher attachment anxiety postpartum and with stress due to loss of employment (Table 2). Other risk factors were not associated with being at risk for PPD (EPDS score ≥ 10).

There was a moderate correlation between Zung's anxiety score before and after giving birth (r = 0.44; p < 0.001). The anxiety score after giving birth on average decreased by 4 points (SD = 6; p < 0.001).

There was a strong correlation between attachment avoidance before and after giving birth (0.71; p < 0.001), and between attachment anxiety before and after giving birth (r = 0.53; p < 0.001). The results suggest that anxiety and anxious attachment style are prone to change after giving birth. This explains the decision of including the before and after birth measures in the regression model separately.

From all the predictors included in the multivariable linear regression model, only education and anxiety after giving birth were positively associated with the depression score after giving birth, when other predictors were controlled for. The EPDS after giving birth was higher in more educated and more anxious nulliparas (Table 3). By the regression model, 60% of variance in the EPDS is explained.

Summary of results

We explored various factors associated with postpartum depression in first-time pregnant women that could represent increased risk for developing postpartum symptoms of depression in first-time pregnant women.

The results of logistic regression models showed that increased risk for developing PPDS was associated with anxiety postpartum, intimate-partner-attachment anxiety postpartum, and elevated stress due to loss of employment or an unsuccessful search for employment in the previous year, when controlled for the prepartum depression score.

However, the results from the multivariable linear regression model demonstrated that only education and anxiety after birth were associated with postpartum depression in nulliparas when other predictors were controlled for. The EPDS after giving birth was higher for more educated and more anxious nulliparas.

Discussion

Multivariable regression analysis showed that the perception of women's experience of childbirth in our sample was not significantly related to higher scores on the EPDS. Of note, women with positive experience of

Table 2 Association between demographic, social and environmental factors and a high EPDS (results of adjusted logistic regression)

	D.score < 10	D.score > =10	aOR ^a (95%CI)	P
Mean age (SD; n)	30.6 (4.1; 131)	31.1 (4; 25)	1 (0.9; 1.2)	0.442
Mean years of education (SD; n)	16.4 (2.3; 131)	16.5 (1.8; 25)	1 (0.8; 1.2)	0.888
Planned pregnancy	109/131 (83.2)	20/25 (80)	0.9 (0.3; 2.9)	0.913
Caesarean section	25/126 (19.8)	7/24 (29.2)	1.8 (0.7; 4.9)	0.251
Positive birth experience	108/130 (83.1)	18/22 (81.8)	0.9 (0.3; 3.1)	0.685
Multigenerational household	26/130 (20)	7/25 (28)	1.4 (0.5; 3.8)	0.517
Constant support from partner	110/125 (88)	21/25 (84)	0.9 (0.3; 3.5)	0.982
Constant support from parents	78/125 (62.4)	16/25 (64)	1.1 (0.5; 2.8)	0.798
Constant support from friends	57/126 (45.2)	9/25 (36)	0.8 (0.3; 1.9)	0.554
High support from co-workers bb	60/124 (48.4)	6/23 (26.1)	0.4 (0.1; 1.1)	0.087
Mean attachment avoidance bb (SD; n)	1.8 (0.7; 121)	2.1 (0.7; 23)	1 (1; 1.1)	0.176
Mean attachment anxiety bb (SD; n)	1.6 (0.5; 119)	2 (0.9; 22)	1.04 (1; 1.08)	0.075
Mean attachment avoidance ab (SD; n)	1.8 (0.7; 114)	2.1 (0.8; 21)	1 (1; 1.1)	0.217
Mean attachment anxiety ab (SD; n)	1.6 (0.6; 122)	2.3 (1.1; 22)	1.1 (1; 1.1)	0.005
Stress – loss of employment	8/125 (6.4)	7/25 (28)	5.5 (1.7; 17.3)	0.004
Stress – family member or close friend's illness or injury	21/126 (16.7)	6/25 (24)	1.4 (0.5; 4.1)	0.500
Stress – own illness or injury	7/124 (5.6)	3/25 (12)	2 (0.5; 8.8)	0.335
Stress – financial difficulties	15/126 (11.9)	4/24 (16.7)	1 (0.3; 3.7)	0.983
Psychiatric help before pregnancy	14/130 (10.8)	2/25 (8)	0.8 (0.2; 4)	0.826
Mean anxiety score bb (SD; n)	32.8 (5.5; 121)	35.6 (6; 23)	1.1 (1; 1.2)	0.109
Mean anxiety score ab (SD; n)	28.3 (4.8; 126)	36.3 (5.7; 24)	1.3 (1.2; 1.5)	< 0.001

^aOR odds ratio, D. score Edinburgh depression score, aOR OR adjusted for baseline depression score

delivery were less likely to drop out of the study. A recent meta-analysis [9] reported that Caesarean sections increased the risk of PPD. However, in our sample, we found no relationship between Caesarean sections and PPD.

The relationship between unplanned pregnancy and postpartum level of depression (higher EPDS score) in our sample was nonsignificant, which is inconsistent with a recent meta-analysis that found the prevalence of perinatal depression to be twice as high in women with an unintended pregnancy [40], but consistent with other longitudinal studies reporting statistically nonsignificant associations between unintended pregnancies and PPD [41, 42].

In contrast with previous studies that reported lack of social support to be a risk factor for PPD [43, 44], we found that the perception of emotional support from a partner, parents, friends or co-workers during pregnancy was not significantly related to higher scores on the EPDS. This finding might also be due to the relatively stable relationships of women who participated in our study. A majority of the participants (87.3%) reported constant support from parents, 43,7% reported constant friends' support and 44,9% reported support from co-workers.

Partner- attachment anxiety postpartum was found to be positively associated with EPDS in the univariate logistic model, which is consistent with the results of other studies; anxious styles were found to be associated with PPDS more frequently than avoidant styles of attachment [13, 14]. The multivariable regression model, however, did not confirm the association of attachment style and PPD.

Elevated stress due to loss of employment or an unsuccessful search for employment in the previous year was associated with higher odds for PPDS in the logistic regression model. In line with the results of our study, a recent study found employment, compared with unemployment, to be significantly associated with a reduced risk for postpartum depression [23]. However, in our study, the association was not confirmed in the multivariable regression model.

Several studies have demonstrated depression and anxiety during pregnancy to be risk factors for PPD [6, 7]. Some studies have even shown anxiety disorders during pregnancy to be a stronger predictor of PPD than depression [45, 46]. In our sample, however, only anxiety postpartum was related to PPDS and the correlation between postpartum anxiety and PPDS was highly

Table 3 Association between demographic, social, environmental and personality variables and EPDS (results of multiple regression analysis)

	Std. B ^a (P)
Age	0.09 (0.348)
Years of education	0.22 (0.029)
Planned pregnancy	-0.04 (0.727)
Caesarean section	0.08 (0.438)
Positive birth experience	0.07 (0.483)
Multigenerational household	0.1 (0.317)
Constant support from partner	-0.15 (0.12)
Constant support from parents	-0.02 (0.88)
Constant support from friends	0.08 (0.482)
High support from co-workers bb	0.09 (0.383)
Attachment avoidance bb	-0.02 (0.907)
Attachment anxiety bb	0.1 (0.482)
Attachment avoidance ab	0.01 (0.964)
Attachment anxiety ab	0.08 (0.561)
Stress – loss of employment	-0.04 (0.654)
Stress – family member or close friend's illness or injury	0 (0.992)
Stress – own illness or injury	-0.04 (0.682)
Stress – financial difficulties	-0.07 (0.538)
Psychiatric help before pregnancy	-0.03 (0.79)
Anxiety score bb	-0.12 (0.34)
Anxiety score ab	0.72 (< 0.001)
Depression score bb	0.1 (0.448)

^aStd. B standardized regression coefficient

significant in both statistical models. One obvious explanation for this could be the overlap of symptoms as well as comorbidity, which is in line with the results of previous research that found anxiety disorders to be highly comorbid with peripartum depression [47]. A recent study reported that 13% of women experienced comorbid depressive symptomatology and anxiety during the first eight weeks postpartum [48, 49]. The more anxious nulliparas in our study also experienced higher levels of depression.

In the current study, the prevalence of being evaluated as being at risk for depression was 17,9% in the third trimester and 16% six weeks after birth. This is in line with, or slightly higher than, what has been reported in previous studies [13, 30], possibly due to the cut-off of EPDS≥10. Similarly, the study on a sample of 449 Slovenian new mothers found that 21.3% of the new mothers had a score of 10 or higher on the EPDS, but the reported study did not exclude multiparas, therefore our results cannot be directly compared [28].

Interestingly, a history of psychiatric help before pregnancy was not related to higher scores on the EPDS. Ten per cent of women in our sample reported getting psychiatric help before pregnancy, yet we found no association with PPD in any of the models, which further emphasizes the importance of searching for less obvious risk factors.

In the current study, higher education was related to higher depression levels, in contrast to studies demonstrating lower education to be associated with PPD [50, 51]. We could speculate that, for highly educated women, taking maternity leave and starting a family represents an additional challenge in their career development and can generate stress that could increase vulnerability to PPDS. Based on this, we might consider the cognitive dissonance of nulliparous women who want an active role in their career. For the highly educated women of our study, pregnancy and the transition to parenthood might have been especially important for their mood level.

Strengths & Limitations

This study has some limitations. Most of the participating women in our study were highly educated first-time pregnant women living in an urban environment which might not be representative of the Slovenian population and the results should probably not be generalized. Possible reasons for these limitations are the use of a university hospital for the study and the inclusion of Slovenian-speaking women only.

The participation rate in our study was approximately 55%, which is in line with similar studies [30]. Of note, nulliparas with positive experience of birth and lower baseline attachment anxiety were less likely to drop out of the study, while women experiencing high stress due to their own physical or mental illness in the last 12 months were less likely to provide complete answers (p = 0.013), which might have led to an underestimation of PPD and factors associated with higher EPDS our sample.

When compared to other nulliparas who gave birth in University Medical Centre (UMC) Ljubljana's Division of Gynaecology and Obstetrics in the year 2014, the women in our sample were more educated and slightly older (statistical significance is not known). The mean years of education in our sample was 16,4 (higher education—university level), while 37.3% of nulliparas that gave birth in 2014 had 16 years of education (university level). The mean age of nulliparas that gave birth in 2014 was 29.35, compared to 30.7 years in our sample.

Our diagnostic assessment focused on mood symptoms, not disorders.

Finally, the sample size of our study was rather small. One of the reasons might be the length of the question-naire and time/schedule limitations of first-time mothers 6 weeks after delivery, which might also account for the

missing data. Future studies should focus on trying to have larger sample sizes.

The strengths of the study include the prospective design, where stress-related and other factors were measured in the last trimester of pregnancy and six weeks after birth, and the contextual framework/design.

Implications

Some practical implications may be taken from the present findings. Knowledge of the relative importance of risk factors associated with PPDS of first-time pregnant women might add to the recognition of vulnerable women and also help develop effective clinical interventions. The peripartum period is a time when most women are in routine contact with health professionals (e.g. midwives). Attention to symptoms of anxiety when assessing peripartum mental health might improve earlier recognition of PPDS and enable early preventive or treatment interventions. Sub-threshold symptom recognition is important in this critical time of early postpartum, when non-pharmacological interventions are of even greater importance. Including questions about experiencing anxiety symptoms in peripartum questionnaires might prove useful in screening for high risk of PPDS.

Conclusions

Our findings demonstrate the importance of anxiety symptoms and higher education level in assessments of nulliparas' mental health. To our knowledge, our study is unique in exploring the contextual understanding of factors, associated with nulliparas' PPDS in a group of first-time pregnant women in Slovenia.

The results of our study show and confirm the results of previous research that anxiety symptoms in the immediate postpartum period are likely to be associated with depressive symptoms in nulliparas. The results also suggest that higher level of education of first-time mothers might not be a protective factor, especially for nulliparas with university level of education.

Attention to the symptoms of anxiety in highly educated nulliparas with seemingly protective psychosocial determinants might add to recognition of vulnerable women in assessment of PPDS.

The logistic regression model showed the association of anxious attachment style and unemployment with increased risk of PPD; however, given that the significant findings from the univariate analysis were not confirmed in the multivariable regression model, we suggest further studies with larger samples.

Abbreviations

ECR-R: Experiences in Close Relationships-Revised; EPDS: The Edinburgh Postpartum Depression Scale; EU: European Union; NMEC: National Medical Ethics Committee of Republic of Slovenia; PPD: Postpartum depression;

PPDS: Postpartum depression symptoms; UMC: University Medical Centre Ljubljana's Division of Gynaecology and Obstetrics

Acknowledgements

Not applicable.

Authors' contributions

PRP led the study, designed and drafted the manuscript. MRM designed the study and critically appraised the manuscript. MVS and TPS helped with data collection. All authors read and approved the final manuscript.

Funding

This project was supported by internal institutional funds.

Availability of data and materials

The data are not publicly available due to their containing information that could compromise research participant privacy/consent. The data are available from the corresponding author on reasonable request.

Ethics approval and consent to participate

The study was approved by the Republic of Slovenia National Medical Ethics Committee (NMEC), protocol No. 92/12/13.

Consent for publication

Not applicable

Competing interests

The authors declare that they have no competing interests.

Author details

¹University Psychiatric Clinic Ljubljana, Studenec 48, 1000 Ljubljana, Slovenia. ²Division of Gynaecology and Obstetrics, University Medical Centre Ljubljana, Zaloška cesta 7, 1000 Ljubljana, Slovenia. ³Faculty of Medicine, University of Ljubljana, Vrazov trg 2, 1000 Ljubljana, Slovenia.

Received: 3 April 2019 Accepted: 9 September 2019 Published online: 18 September 2019

References

- Banti S, Mauri M, Oppo A, Borri C, Rambelli C, Ramacciotti D, et al. From the third month of pregnancy to 1 year postpartum. Prevalence, incidence, recurrence, and new onset of depression. Results from the Perinatal Depression–Research & Screening Unit study. Compr Psychiatry. 2011;52(4): 343–51. https://doi.org/10.1016/j.comppsych.2010.08.003.
- Glover V. Maternal depression, anxiety and stress during pregnancy and child outcome; what needs to be done. Best Pract Res Clin Obstet Gynaecol. 2014;28(1):25–35. https://doi.org/10.1016/j.bpobgyn.2013.08.017.
- Dunkel Schetter C, Tanner L. Anxiety, depression and stress in pregnancy. Curr Opin Psychiatry. 2012;25(2):141–8. https://doi.org/10.1097/yco. 0b013e3283503680.
- Hübner-Liebermann B, Hausner H, Wittmann M. Recognizing and treating Peripartum depression. Dtsch Arztbl Int. 2012;109(24):419–24. https://doi. org/10.3238/arztebl.2012.0419.
- Giardinelli L, Innocenti A, Benni L, Stefanini M, Lino G, Lunardi C, et al. Depression and anxiety in perinatal period: prevalence and risk factors in an Italian sample. Arch Wom Ment Health. 2011;15(1):21–30. https://doi.org/10. 1007/s00737-011-0249-8.
- Robertson E, Grace S, Wallington T, Stewart D. Antenatal risk factors for postpartum depression: a synthesis of recent literature. Gen Hosp Psychiatry. 2004;26(4):289–95. https://doi.org/10.1016/j.genhosppsych.2004.02.006.
- Goodman JH, Chenausky KL, Freeman MP. Anxiety disorders during pregnancy. J Clin Psychiatry. 2014;75(10):1153–84. https://doi.org/10.4088/ jcp.14r09035.
- Darcy JM, Grzywacz JG, Stephens RL, Leng I, Clinch CR, Arcury TA. Maternal depressive symptomatology: 16-month follow-up of infant and maternal health-related quality of life. J Am Board Fam Med. 2011;24(3):249–57. https://doi.org/10.3122/jabfm.2011.03.100201.
- Xu H, Ding Y, Ma Y, Xin X, Zhang D. Cesarean section and risk of postpartum depression: a meta-analysis. J Psychosom Res. 2017;97:118–26. https://doi.org/10.1016/j.jpsychores.2017.04.016.

- Rini C, Schetter C, Hobel C, Glynn L, Sandman C. Effective social support: antecedents and consequences of partner support during pregnancy. Pers Relatsh. 2006;13(2):207–29. https://doi.org/10.1111/j.1475-6811.2006.00114.x.
- Pepping C, Halford W. Attachment and relationship satisfaction in expectant first-time parents: the mediating role of relationship enhancing behaviors. J Res Pers. 2012;46(6):770–4. https://doi.org/10.1016/j.jrp.2012.08.005.
- Lee J, Koo H. The relationship between adult attachment and depression in Korean mothers during the first 2 years postpartum: a moderated mediation model of self-esteem and maternal efficacy. Pers Individ Dif. 2015;79:50–6. https://doi.org/10.1016/j.paid.2015.01.021.
- Warfa N, Harper M, Nicolais G, Bhui K. Adult attachment style as a risk factor for maternal postnatal depression: a systematic review. BMC Psychol. 2014; 2(1):56. https://doi.org/10.1186/s40359-014-0056-x.
- Bifulco A, Figueiredo B, Guedeney N, Gorman L, Hayes S, Muzik M, et al. Maternal attachment style and depression associated with childbirth: preliminary results from a European and US cross-cultural study. Br J Psychiatry. 2004;184(46):31–7. https://doi.org/10.1192/bjp.184.46.s31.
- Ikeda M, Hayashi M, Kamibeppu K. The relationship between attachment style and postpartum depression. Attach Hum Dev. 2014;16(6):557–72. https://doi.org/10.1080/14616734.2014.941884.
- Monk C, Leight K, Fang Y. The relationship between women's attachment style and perinatal mood disturbance: implications for screening and treatment. Arch Womens Ment Health. 2008;11(2):117–29. https://doi.org/10. 1007/s00737-008-0005-x.
- Robakis TK, Williams KE, Crowe S, Lin KW, Gannon J, Rasgon NL. Maternal attachment insecurity is a potent predictor of depressive symptoms in the early postnatal period. J Affect Disord. 2016;190:623–31. https://doi.org/10. 1016/j.jad.2015.09.067.
- Simard V, Moss E, Pascuzzo K. Early maladaptive schemas and child and adult attachment: a 15-year longitudinal study. Psychol Psychother. 2011; 84(4):349–66. https://doi.org/10.1111/j.2044-8341.2010.02009.x.
- Roisman G, Fraley R. A behavior-genetic study of parenting quality, infant attachment security, and their covariation in a nationally representative sample. Dev Psycho. 2008;44(3):831–9. https://doi.org/10.1037/0012-1649.44.3.831.
- Fearon RM, Roisman GI. Attachment theory: progress and future directions. Curr Opin Psychol. 2017;15:131–6. https://doi.org/10.1016/j.copsyc.2017.03.002.
- Selboe ST, Skogås AK. Working fulltime throughout pregnancy the Norwegian women's perspectives. A qualitative approach. Midwifery. 2017; 50:193–200. https://doi.org/10.1016/j.midw.2017.04.012.
- Gjerdingen D, McGovern P, Attanasio L, Johnson P, Kozhimannil KB. Maternal depressive symptoms, employment and social support. J Am Board Fam Med. 2014;27(1):87–96. https://doi.org/10.3122/jabfm.2014.01. 130126.
- Miyake Y, Tanaka K, Sasaki S, Hirota Y. Employment, income, and education and risk of postpartum depression: the Osaka maternal and child health study. J Affect Disord. 2011;130(1–2):133–7. https://doi.org/10.1016/j.jad. 2010.10.024.
- Gender statistics Statistics Explained [Internet]. [place unknown]. Ec.europa. eu. 2018. [updated 2018 Feb 2018; cited 2018 Sep 24]. Available from: https://ec.europa.eu/eurostat/statistics-explained/index.php/Gender_ statistics
- Räisänen S, Lehto SM, Nielsen HS, Gissler M, Kramer MR, Heinonen S. Fear of childbirth predicts postpartum depression: a population-based analysis of 511 422 singleton births in Finland. BMJ Open. 2013;3(11):e004047. https://doi.org/10.1136/bmjopen-2013-004047.
- Iwata H, Mori E, Sakajo A, Aoki K, Maehara K, Tamakoshi K. Prevalence of postpartum depressive symptoms during the first 6 months postpartum: association with maternal age and parity. J Affect Disord. 2016;203:227–32.
- 27. O'Connell MA. Preventing, recognising, and responding to fear of childbirth and birth trauma. BMJ. 2019;365:12279.
- Koprivnik P, Plemenitaš A. Pojavnost depresij v poporodnem obdobju v slovenskem vzorcu. Med Razgl. 2005;44:139–45.
- Dennis CL, Dowswell T. Psychosocial and psychological interventions for preventing postpartum depression. Cochrane Database Syst Rev. 2013;(2). https://doi.org/10.1002/14651858.CD001134.pub3.
- Sylvén SM, Thomopoulos TP, Kollia N, Jonsson M, Skalkidou A. Correlates of postpartum depression in first time mothers without previous psychiatric contact. Eur Psychiatry. 2017;40:4–12. https://doi.org/10.1016/j.eurpsy.2016.07.003.
- Yim IS, Tanner Stapleton LR, Guardino CM, Hahn-Holbrook J, Dunkel Schetter C. Biological and psychosocial predictors of postpartum depression: systematic review and call for integration. Annu Rev Clin

- Psychol. 2015;11(1):99–137. https://doi.org/10.1146/annurev-clinpsy-101414-020426
- 32. O'Brien BC, Harris IB, Beckman TJ, Reed DA, Cook DA. Standards for reporting qualitative research: a synthesis of recommendations. Acad Med. 2014;89(9):1245–51.
- Fraley RC, Waller NG, Brennan KA. An item response theory analysis of selfreport measures of adult attachment. J Pers Soc Psychol. 2000;78(2):350–65. https://doi.org/10.1037//0022-3514.78.2.350.
- Cox JL, Holden J, Henshaw C. Perinatal mental health: the EPDS manual. 2nd ed. London: RCPsych Publications; 2014.
- Cox JL, Chapman G, Murray D, Jones P. Validation of the Edinburgh postnatal depression scale (EPDS) in non-postnatal women. J Affect Disord. 1996;39:185–9. https://doi.org/10.1016/0165-0327(96)00008-0.
- Koprivnik P, Plemenitaš, A. Screening for postpartum depression why is it not implemented in Slovenia? Zdrav Vestn. [Internet]. 2009 [cited 2018 Sep 24]. 78(5): 261–264. Available from: http://www.dlib.si/?URN=URN:NBN:SI: DOC-NHG94LAA
- 37. Zung WW. A rating instrument for anxiety disorders. Psychosomatics. 1971; 12(6):371–9. https://doi.org/10.1016/S0033-3182(71)71479-0.
- Peduzzi P, Concato J, Kemper E, Holford TR, Fein-stein AR. A simulation study of the number of events per variable in logistic regression analysis. J Clin Epidemiol. 1996;49:1373–9.
- Agresti A. In: Hoboken NJ, editor. An introduction to categorical data analysis: Wiley; 2007.
- Abajobir AA, Maravilla JC, Alati R, Najman JM. A systematic review and meta-analysis of the association between unintended pregnancy and perinatal depression. J Affect Disord. 2016;192:56–63. https://doi.org/10. 1016/j.jad.2015.12.008.
- Mercier RJ, Garrett J, Thorp J, Siega-Riz AM. Pregnancy intention and postpartum depression: secondary data analysis from a prospective cohort. BJOG. 2013;120(9):1116–22. https://doi.org/10.1111/1471-0528.12255.
- 42. Christensen AL, Stuart EA, Perry DF, Le HN. Unintended pregnancy and perinatal depression trajectories in low-income, high-risk Hispanic immigrants. Prev Sci. 2011;12(3):289–99. https://doi.org/10.1007/s11121-011-0213-x.
- Xie RH, He G, Koszycki D, Walker M, Wen SW. Fetal sex, social support, and postpartum depression. Can J Psychiatr. 2009;54(11):750–6. https://doi.org/ 10.1177/070674370905401105.
- Shapiro GD, Fraser WD. Review: psychosocial and psychological interventions reduce postpartum depression. Ann Intern Med. 2013;159(4): JC8. https://doi.org/10.7326/0003-4819-159-4-201308200-02008.
- Della Vedova AM, Ducceschi B, Cesana BM, Imbasciati A. Maternal bonding and risk of depression in late pregnancy: a survey of Italian nulliparous women. J Reprod Infant Psychol. 2011;29:208–22. https://doi.org/10.1080/ 02646838.2011.592973.
- Lee AM, Lam SK, Sze Mun Lau SM, Chong CS, Chui HW, Fong DY. Prevalence, course, and risk factors for antenatal anxiety and depression. Obstet Gynecol. 2007;110(5):1102–12. https://doi.org/10.1097/01.AOG. 0000287065.59491.70.
- Ross LE, Evans SE, Sellers EM, Romach MK. Measurement issues in postpartum depression part 1: anxiety as a feature of postpartum depression. Arch Womens Ment Health. 2003;6(1):51–7. https://doi.org/10. 1007/s00737-002-0155-1.
- Falah-Hassani K, Shiri R, Dennis CL. Prevalence and risk factors for comorbid postpartum depressive symptomatology and anxiety. J Affect Disord. 2016; 198:142–7. https://doi.org/10.1016/j.jad.2016.03.010.
- Kirchmann H, Steyer R, Mayer A, Joraschky P, Schreiber-Willnow K, Strauss B. Effects of adult inpatient group psychotherapy on attachment characteristics: an observational study comparing routine care to an untreated comparison group. Psychother Res. 2012;22(1):95–114. https://doi. org/10.1080/10503307.2011.626807.
- Pham D, Cormick G, Amyx MM, Gibbons L, Doty M, Brown A, Norwood A, Daray FM, Althabe F, Belizán JM. Factors associated with postpartum depression in women from low socioeconomic level in Argentina: a hierarchical model approach. J Affect Disord. 2018;227:731–8.
- Bad HE, Alhamdan N, Ajaj A, Alali F. Postpartum depression and health related quality of life: a necessary assessment. Int J Fam Commun Med. 2017;1(1):00005.

Publisher's Note

Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.